Author's response to reviews

Title: Fetal size in the second trimester is associated with the duration of pregnancy, small fetuses having longer pregnancies.

Authors:

Synnove L Johnsen (synnove.johnsen@helse-bergen.no)
Tom Wilsgaard (Tom.Wilsgaard@ism.uit.no)
Svein Rasmussen (svein.rasmussen@mfr.uib.no)
Mark A Hanson (M.Hanson@soton.ac.uk)
Keith M Godfrey (kmg@mrc.soton.ac.uk)
Torvid Kiserud (torvid.kiserud@kk.uib.no)

Version: 3 Date: 19 June 2008

Author's response to reviews: see over
Dear Chief Editor,

We are very pleased to hear that you have accepted our manuscript 1074165597170903 for publication in BMC Pregnancy and Childbirth.

- The title is now changed according to the suggestion from the editorial board member:
  “Fetal size in the second trimester is associated with the duration of pregnancy, small fetuses having longer pregnancies”. We have expanded the phrasing to more clearly reflect the findings of our study.

- Concerning Referee 2’s point on the HC chart used to adjust for the difference between LMP and HC-estimated age in the present study, this chart is very much in line with the British HC-chart of Altman and Chitty (1, 2). A reanalysis using a second chart would be extremely unlikely to alter the analysis and conclusion.

- We have ensured that the format is in line with the manuscript formatting checklist.

- Please find attached the Revised Version of our manuscript.

References:

